Author's response to reviews

Title: Delayed Diagnosis of Anterior Subluxation of C7 vertebrae in a Post-Traumatic 35 years old with unusual neurological pattern - a case report

Authors:

Maryam Sanaullah (mariamtariq27@hotmail.com)
Abdul Sattar Mohammad Hashim (2neurospi@cyber.net.pk)
Ayesha Sundus (3avesha_sundus@hotmail.com)
Sanaullah Bashir (sanaullahbashir@gmail.com)
Maheen Rehman (onlymaheen@hotmail.com)

Version: 2 Date: 12 May 2012

Author's response to reviews: see over
COVERING LETTER

The editor,

Journal of Medical Case Report,

Subject: Submission of revised manuscript

Dear Editor,

I would like to submit the following manuscript entitled as “Delayed Diagnosis of Anterior Subluxation of C7 vertebrae in a Post-Traumatic 35 years old with unusual neurological pattern – a case report” after revision.

Following revisions have been made:

1. Study design is included in title.
2. Patient ethnicity has been mentioned in both abstract and main manuscript.
3. Consent statement has been revised.
4. Competing interest statement has been reformed.
5. Author’s contribution has been revised.

“Written informed consent was obtained from the patient for publication of this manuscript and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.”

Type of Manuscript: Case Report.

Maryam Sanaullah